

# CMS-EMS Center for Mathematical Studies in Economics And Management Science

Discussion Paper #1555

# VIGILANT MEASURES OF RISK AND THE DEMAND FOR CONTINGENT CLAIMS

MARIO GHOSSOUB

UNIVERSITÉ DE MONTRÉAL

October 22, 2012

JEL Classification: C02, D81, D89, G11

*Keywords*: Contingent Claims, Heterogeneous Beliefs, Choquet Integral, Vigilance, Monotone Likelihood Ratio





# VIGILANT MEASURES OF RISK AND THE DEMAND FOR CONTINGENT CLAIMS

## MARIO GHOSSOUB UNIVERSITÉ DE MONTRÉAL

THIS DRAFT: OCTOBER 22, 2012

LATEST DRAFT: https://sites.google.com/site/marioghossoub

ABSTRACT. I examine a class of utility maximization problems with a not necessarily law-invariant utility, and with a not necessarily law-invariant risk measure constraint. The objective function is an integral of some function  $\mathcal U$  with respect to some probability measure P, and the constraint set contains some risk measure constraint which is not necessarily P-law-invariant. This introduces some heterogeneity in the perception of uncertainty. The primitive  $\mathcal U$  is a function of some given underlying random variable X and of a contingent claim Y on X. Many problems in economic theory and financial theory can be formulated in this manner, when a heterogeneity in the perception of uncertainty is introduced. Under a consistency requirement on the risk measure that will be called Vigilance, supermodularity of the primitive  $\mathcal U$  is sufficient for the existence of optimal continent claims, and for these optimal claims to be comonotonic with the underlying random variable X. Vigilance is satisfied by a large class of risk measures, including all distortion risk measures. An explicit characterization of an optimal contingent claim is also provided in the case where the risk measure is a convex distortion risk measure.

#### 1. Introduction

Problems of "utility" maximization in financial theory are not only some of the most fundamental issues, but they have also been recently re-examined in general abstract settings, hence allowing for various applications. See, for instance, Carlier and Dana [9, 10], Cvitanić, Schachermayer, and Wang [14], Dana [15], Föllmer and Schied [21], Hugonnier and Kramkov [31], Kramkov and Schachermayer [36, 37], Owen and Žitković [45], Schachermayer [51], Schied [52, 53, 54, 55], and Schied and Wu [56], to cite only a few. The generality of such problems

Key words and phrases. Contingent Claims, Heterogeneous Beliefs, Choquet Integral, Vigilance, Monotone Likelihood Ratio.

JEL Classification: C02, D81, D89, G11.

<sup>2010</sup> Mathematics Subject Classification: 91G10, 91G20, 91B42.

I am grateful to Alexander Schied and Costis Skiadas for their comments and suggestions. I thank Max Amarante, Hirbod Assa, Marc Henry, Andrew Heunis, Adam Kolkiewicz, and Marcus Pivato for their comments on earlier drafts. I also thank audiences at the University of British Columbia, the University of Michigan, l'Université de Montréal, the Montreal Seminar of Actuarial and Financial Mathematics, HEC Montréal, and RUD 2012 at Northwestern University. Any errors or omissions are entirely mine. Part of this research was done while I was visiting the Center of Mathematical Studies in Economic and Management Science (CMS-EMS), at the Kellogg School of Management – Northwestern University, which I thank for its hospitality. Financial support from the Natural Sciences and Engineering Research Council of Canada and from the Social Sciences and Humanities Research Council of Canada is gratefully acknowledged.

allows for numerous interpretations, depending on the particular consumer demand problem at hand. By far the most common application of these problems is the – by now folkloric – problem of portfolio choice under uncertainty. The road paved by the seminal contributions of Merton [40, 41] has been often walked on, and the literature on portfolio choice is far too large for us to give a meaningful review here. Nevertheless, some of the recent contributions include He and Zhou [29], Jin and Zhou [33, 34, 35], and Zhang, Jin, and Zhou [62], for instance. These authors use a novel approach to utility maximization, based on a quantile reformulation of optimization problems, whereby problems involving a given choice variable are replaced by problems involving this variable's quantile function. Such quantile problems often turn out to be easier to handle and better behaved.

In this paper I use similar quantile techniques to show the existence of monotonic solutions to a general utility maximization problem, with a non-law-invariant indirect utility and a non-law-invariant risk measure constraint. This problem can be seen as a problem of demand for contingent claims under uncertainty, and it is an abstraction of many common problems in economic theory that were hitherto only considered in a framework of complete homogeneity of beliefs about the realizations of an underlying uncertainty. It can be formulated as

(1.1) 
$$\sup_{Y \in \Theta} V(X,Y) := \int \mathcal{U}(X,Y) \ dP$$

where X is a given random variable on a probability space  $(S, \Sigma, P)$ ,  $B(\Sigma)$  is the linear space of all bounded and  $\Sigma$ -measurable functions on S,  $\Theta \subset B(\Sigma)$  is a given non-empty constraint set, and  $\mathcal{U}(X,Y)$  is bounded and  $\Sigma$ -measurable for each  $Y \in \Theta$ . The objective function V(X,Y) is understood to be a decision maker's (DM) indirect utility function, that is, V(X,Y) represents the DM's expected utility of terminal wealth.

1.1. Objective vs. Subjective Uncertainty. When the constraint set  $\Theta$  of Problem (1.1) contains, for example, another party's individual rationality constraint (participation constraint), one can distinguish between two types of problems, depending on how the underlying uncertainty is perceived by both parties: (i) either both parties agree on the distribution of this uncertainty (which will hence be induced by the probability measure P), or (ii) they have different perceptions of such randomness. The first type of problem is one where uncertainty can be called *homogeneous*, whereas the second type is a problem in which uncertainty can be referred to as being *heterogeneous*.

Surprisingly, the literature is mostly silent on problems of the form (1.1) where the uncertainty is heterogeneous, whereas problems of the form (1.1) with homogeneous uncertainty are abundant. For example, the vast majority of problems of optimal insurance design, or demand for insurance coverage are based on the classical formulation of Arrow [3], Borch [8], and Raviv [48], and are usually stated as a problem of the form (1.1) with homogeneous uncertainty. That is, both the insurer and the insured share the same beliefs about the realizations of some underlying insurable loss random variable X. In these problems, monotonicity of an optimal insurance contract Y is typically desired because such contracts can avoid ex-post moral hazard that might arise from a voluntary downward misrepresentation of the loss by the insured.

Problems of debt contracting between investors (lenders) and entrepreneurs (borrowers), such as the ones studied by Gale and Hellwig [22], Townsend [59], or Williamson [61], are also usually stated as a problem of the form (1.1) with homogeneous uncertainty. In this case, a contract specifies the repayment Y that the borrower makes to the lender as a function of the (uncertain)

return X on the project being financed. The monotonicity of an optimal contract as a function of the return on investment is a coveted feature since such contracts will be de facto truthtelling, and will avoid any misrepresentation of the profitability of the project by the borrower.

Principal-agent problems have also been traditionally stated as problems of the form (1.1) with homogeneous uncertainty, as in the work of Grossman and Hart [27], Holmstrom [30], Mirrlees [44], Page [46], or Rogerson [49], for instance. In that setting, a contract specifies the wage Y that an agent receives from the principal, as a function of the (uncertain) outcome, or output X that occurs as a result of the agent's activity. Since the work of Rogerson [49], monotonicity of the optimal wage contract in the observed output is usually sought after<sup>1</sup>.

Numerous other problems can be formulated as in (1.1), such as problems of demand for financial securities given a pricing or budgeting constraint. In such problems, monotonicity properties of optimal claims were studied first by Dybvig [18, 19]. Problem (1.1) can also be interpreted as a Neyman-Pearson type problem. In these problems, monotonicity properties of optimal solutions were discussed by Schied [52], for instance. Whatever the nature of the problem might be, it is interesting to examine under what conditions an optimal choice of the choice variable Y is monotone in the underlying variable X, and the theory of monotone comparative statics has usually been very fruitful in answering similar questions of monotonicity of solutions. However, it is of no help here, as the next section argues.

1.2. The Theory of Monotone Comparative Statics and its Limitations. The importance of monotone comparative statics analyses in economic theory is well-understood. One can even say that at the core of the motivation behind a sizeable collection of problems in economic theory, very often lies the question of whether or not a quantity is a monotone function of a parameter, or whether a variable output changes monotonically with a variable input. This is even more so significant if the monotonicity of an optimal such output as a function of an input parameter is desired, and indeed, monotone comparative statics techniques have proven to be very fruitful (see [58, 60]). Such techniques can be, and have been used in consumer theory, theory of production, portfolio choice theory, financial economics, and contract theory to answer some basic and intuitive questions.

The theory of monotone comparative statics is typically concerned with the behavior of a solution to a given optimization problem when a primitive of the problem changes. Specifically, let  $(L, \ge_L)$  be a lattice,  $B \subseteq L$  a choice set,  $(T, \ge_T)$  a partially ordered set interpreted as a set of parameters, and  $f: L \times T \to \mathbb{R}$  a given objective function. For the problem of choosing an  $x \in B$  that maximizes the objective function given a value t of the parameter, the chief concern is the isotonicity of an optimal choice  $x^*(t)$  of x as a function of t, that is,

$$(1.2) t_1 \geqslant_T t_2 \implies x^*(t_1) \geqslant_L x^*(t_2)$$

The classical theory of monotone comparative statics [42, 43, 57, 58] seeks conditions on the function f that guarantee that eq. (1.2) holds.

Athey [5] examined a problem of monotone comparative statics in the presence of uncertainty, where the objective function is an integral of some function with respect to some measure. Specifically, let  $(L, \ge_L)$  be a lattice,  $B \subseteq L$  a choice set,  $S = \prod_{i=1}^m S_i$  with  $S_i \subseteq \mathbb{R}$  for  $i = 1, \ldots, m$ ,  $\Theta \subseteq \mathbb{R}$  a set of parameters,  $\mu$  a finite nonnegative product measure on S, and u:

<sup>&</sup>lt;sup>1</sup>For various reasons that are beyond the scope of this paper.

 $L \times S \to \mathbb{R}$  and  $\psi : S \times \Theta \to \mathbb{R}$  given bounded measurable functions. Define the objective function  $\Phi : L \times \Theta \to \mathbb{R}$  by

$$\Phi(x,\theta) = \int_{S} u(x,s) \psi(s,\theta) d\mu(s)$$

For the problem of choosing an  $x \in B$  that maximizes the objective function given a value  $\theta$  of the parameter, the problem of monotone comparative statics in this situation of uncertainty is to find conditions on the primitives u and  $\psi$  so that an optimal choice  $x^*(\theta)$  of x is a nondecreasing function of  $\theta$ , that is,

$$\theta_1 \geqslant \theta_2 \implies x^*(\theta_1) \geqslant_L x^*(\theta_2)$$

In particular, in both situations of certainty and uncertainty, the interest is in the variation of the optimal solution with respect to the lattice order  $\geqslant_L$  on L, given a variation of the parameter  $(t \text{ or } \theta, \text{ respectively})$  in the order on the parameter set  $(\geqslant_T \text{ or the usual order on } \mathbb{R}, \text{ respectively})$ . Often, however, these notions of order are too strong for the problem under consideration. For instance, in problems of the form (1.1), conditions on the primitive  $\mathcal{U}$  for the optimal choice  $Y^*$  of Y to be monotone in X are desired. Specifically, under what conditions on  $\mathcal{U}$  does one have that for all  $s, s' \in S$ ,  $X(s) \geqslant X(s') \Rightarrow Y^*(s) \geqslant Y^*(s')$ ? The classical techniques of monotone comparative statics are of no help in these situations since the lattice order on  $B(\Sigma)$  is not adequate here. This order  $\geqslant_L$  on  $L = B(\Sigma)$  is defined by

$$Y_1 \geqslant_L Y_2$$
 if and only if  $Y_1(s) \geqslant Y_2(s)$ , for all  $s \in S$ 

1.3. A Class of Demand Problems. Since the seminal contribution of Artzner et al. [4] and Delbaen [16], the literature on risk measures has grown exponentially<sup>2</sup>. This motivated an active area of research where many classical problems arising in financial theory were re-examined using general risk measures beyond the classical dispersion measures such as the variance<sup>3</sup>. The problem that will be examined in this paper can be seen as an abstraction of problems of this sort, and it takes the form

(1.3) 
$$\sup_{Y=I \circ X} \left\{ \int \mathcal{U}(X,Y) \ dP \ \middle| \ 0 \leqslant Y \leqslant X, \ \rho(Y) \leqslant R \right\}$$

where X is a given underlying uncertainty on the measurable space  $(S, \Sigma, P)$ ,  $Y = I \circ X$  is a claim contingent on this uncertainty,  $\int \mathcal{U}(X, I \circ X) \ dP$  is a DM's expected utility of wealth with respect to the probability measure  $P, \rho : B^+(\Sigma) \to \mathbb{R}$  is a given risk measure, and  $R \in \mathbb{R}$  is fixed.

The first constraint is standard in many problems in economic theory. In the insurance framework [3, 48], this constraint says that an indemnity is nonnegative and cannot exceed the loss itself. In a framework of debt contracting [22], this constraint is a limited liability constraint. The second constraint is simply a risk measure constraint, but it can be interpreted as a budget constraint, or a pricing constraint. In some situations, this constraint can also be seen as another party's individual rationality constraint (participation constraint). For instance, in problems of insurance demand, the constraint  $\rho(Y) \leq R$  would be a "premium constraint" of the form  $\int Y dP \leq \Pi$ , which is nothing more than a rephrasing of the (risk-neutral) insurer's participation constraint.

<sup>&</sup>lt;sup>2</sup>See, for instance, Föllmer and Schied [21] for a review.

<sup>&</sup>lt;sup>3</sup>See, for instance, Balbás [6] for a brief review.

In Problem (1.3), the objective function (indirect utility function) is not necessarily law-invariant with respect to P (Definition 2.6). This is an important point of departure from the recent literature on utility maximization, such as in Carlier and Dana [9, 10] or Schied [52]. Moreover, in the constraint set of Problem (1.3), the mapping  $\rho: B^+(\Sigma) \to \mathbb{R}$  need not be law-invariant with respect to P. When  $\rho$  is not law-invariant with respect to P, this creates some heterogeneity in the perception of the uncertainty X, and poses some important mathematical complications. For instance, in the insurance framework, it might be that the DM and the insurer assign different "distributions" to the underlying uncertainty. The latter problem has been examined by Ghossoub [23, 25, 26].

This paper's main result (Theorem 3.1) is that when the risk measure  $\rho$  satisfies a property that will be called Vigilance (Definition 2.7) and a continuity property that will be called the  $Weak\ DC$ -Property (Definition 2.5), supermodularity of the function  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  (Definitions 2.3 and B.5) is sufficient for an optimal choice of  $Y = I \circ X$  to be a nondecreasing function of the underlying uncertainty X. Roughly speaking, vigilance of the risk measure  $\rho$  can be understood as a (weak) preference for comonotonicity with X (Definition 2.2), on the collection of all functions that are identically distributed for the probability measure P. Given two elements  $Y_1$  and  $Y_2$  of  $B^+$  ( $\Sigma$ ) that have the same distribution with respect to the probability measure P, vigilance of a risk measure  $\rho: B^+$  ( $\Sigma$ )  $\to \mathbb{R}$  means that if any one of  $Y_1$  or  $Y_2$  is a nondecreasing function of X, it will assigned a lower value by  $\rho$  than the other function, and it will hence be seen as less risky. The Weak DC-Property of an operator roughly means that the operator preserves dominated convergence. This property is satisfied by a large class of operators on  $B^+$  ( $\Sigma$ ), such as the Lebesgue integral or the Choquet integral (Definition A.3).

The idea of vigilance introduced here is an extension of the notion of vigilant beliefs introduced by Ghossoub [23, 25, 26] in the insurance framework under Bayesian uncertainty, and extended by Amarante, Ghossoub, and Phelps [2] to the case of Knightian uncertainty (Ambiguity). Ghossoub [23, 25, 26] re-examines the classical Arrow-Borch-Raviv model of insurance demand, but allowing for belief heterogeneity between the insurer and the insurance buyer. He shows that if the insurer's subjective probability measure on the state space satisfies a property called vigilance, then an optimal contract for the insurance buyer takes the form of what the author calls a generalized deductible contract. Moreover, he characterizes the collection of all optimal contracts in terms of their distribution for the insurance buyer. In Ghossoub's [23, 25, 26] setting, where X is the insurable loss random variable, P is the insurance buyer's (subjective) probability measure on the state space, and Q is the insurer's (subjective) probability measure on the state space, Q is said to be (P,X)-vigilant if for any two insurance contracts  $I_1(X)$  and  $I_2(X)$  that are identically distributed under P, and such that  $I_2$  is a nondecreasing function<sup>4</sup>, it follows that  $I_2(X)$  is preferred by the insurer to  $I_1(X)$ . Ghossoub [23, 25, 26] argues that (P,X)-vigilance of Q indicates a kind of credibility that the insurer assigns to the insurance buyer's assessment of the riskiness of the insurance contracts being compared, and he shows that vigilance is a (strictly) weaker assumption than that of a monotone likelihood ratio, when the latter can be defined<sup>5</sup>. In this paper, the definition of vigilance is extended from the notion of vigilant beliefs to the concept of a vigilant real-valued mapping  $\rho$  on the collection of functions Y over which a decision maker (DM) has a given preference. When  $\rho(Y) = \int Y dP$ , one will

<sup>&</sup>lt;sup>4</sup>So that  $I_2(X)$  and X are comonotonic (Definition 2.2).

<sup>&</sup>lt;sup>5</sup>That is, when the measure P and Q are such that the laws  $P \circ X^{-1}$  and  $Q \circ X^{-1}$  have densities (with respect to Lebesgue measure), so that a likelihood ratio can be defined as the ratio of these densities.

recover Ghossoub's [23, 25, 26] definition of vigilant beliefs as a special case of the definition of vigilance given here (Definition 2.7).

1.4. Utility Maximization with a Convex Distortion Risk Measure. As an illustration, I consider a utility maximization problem under a risk measure constraint, where the risk measure is a convex distortion risk measure. These risk measures are automatically vigilant and possess the Weak DC Property. Our Theorem 3.1 then yields the existence of optimal contingent claims that are nondecreasing functions of the underlying random variable X. Using a quantile reformulation procedure, I show that under mild regularity conditions on the DM's utility function, an optimal contingent claim has an explicit characterization, and a crisp and tractable form.

Specifically, when  $\mathcal{U}(X,Y) := u(W_0 + Y - X)$ , for some utility function u satisfying some mild regularity conditions, and when the risk measure  $\rho$  is a Choquet integral of the form  $\rho(Y) = \hat{\int} Y dT \circ P$ , for some convex distortion function T, then an optimal contingent claim takes the form

(1.4) 
$$\mathcal{Y}^* = \max \left[ 0, \min \left\{ F_X^{-1}(U), (u')^{-1} \left( \lambda^* T'(1-U) \right) + F_X^{-1}(U) - W_0 \right\} \right]$$

where  $\lambda^*$  is chosen so that  $\int T'(1-U)\mathcal{Y}^* dP = R$ , U is a uniformly distributed random variable on (0,1), and  $F_X^{-1}$  denotes the quantile function of X (defined as in equation (4.1)).

1.5. **Outline.** The rest of this paper is organized as follows. Section 2 introduces some preliminary notation and definitions. Section 3 presents this paper's main result, which gives sufficient conditions for the existence of comonotonic optimal continent claims. Section 4 gives a characterization of comonotonic optimal contingent claims in terms of quantile functions, when the risk measure is a distortion risk measure. Section 5 examines a special case of demand for contingent claims under a convex risk measure, and gives an explicit characterization of a comonotonic optimal continent claim in this case. Finally, Section 6 concludes. Some proofs and some related analysis can be found in the Appendices.

#### 2. Preliminaries and Notation

Let S denote the set of states of the world, and suppose that  $\mathcal{G}$  is a  $\sigma$ -algebra of subsets of S, called events. Denote by  $B(\mathcal{G})$  the supnorm-normed Banach space of all bounded,  $\mathbb{R}$ -valued and  $\mathcal{G}$ -measurable functions on  $(S,\mathcal{G})$ , and denote by  $B^+(\mathcal{G})$  the collection of all  $\mathbb{R}^+$ -valued elements of  $B(\mathcal{G})$ . For any  $f \in B(\mathcal{G})$ , the supnorm of f is given by  $||f||_{sup} := \sup\{|f(s)| : s \in S\} < +\infty$ . For  $C \subseteq S$ , denote by  $\mathbf{1}_C$  the indicator function of C. For any  $A \subseteq S$  and for any  $B \subseteq A$ , denote by  $A \setminus B$  the complement of B in A.

For any  $f \in B(\mathcal{G})$ , denote by  $\sigma\{f\}$  the  $\sigma$ -algebra of subsets of S generated by f, and denote by  $B(\sigma\{f\})$  the linear space of all bounded,  $\mathbb{R}$ -valued and  $\sigma\{f\}$ -measurable functions on  $(S,\mathcal{G})$ . Then by Doob's measurability theorem [1, Theorem 4.41], for any  $g \in B(\sigma\{f\})$  there exists a Borel-measurable map  $\zeta : \mathbb{R} \to \mathbb{R}$  such that  $g = \zeta \circ f$ . Denote by  $B^+(\sigma\{f\})$  the cone of nonnegative elements of  $B(\sigma\{f\})$ .

**Definition 2.1.** A finite nonnegative measure  $\eta$  on a measurable space  $(\Omega, \mathcal{A})$  is said to be nonatomic if for any  $A \in \mathcal{A}$  with  $\eta(A) > 0$ , there is some  $B \in \mathcal{A}$  such that  $B \subsetneq A$  and  $0 < \eta(B) < \eta(A)$ .

For any  $f \in B(\mathcal{G})$ , if  $\mathcal{A}$  is any sub- $\sigma$ -algebra of  $\mathcal{G}$  such that  $\sigma\{f\} \subseteq \mathcal{A}$ , and if P is any probability measure on the measurable space  $(S, \mathcal{A})$ , it will be said that f is a continuous random variable for P when the law  $P \circ f^{-1}$  of f is a nonatomic Borel probability measure.

**Definition 2.2.** Two functions  $Y_1, Y_2 \in B(\mathcal{G})$  are said to be comonotonic if

$$\left[Y_{1}\left(s\right)-Y_{1}\left(s'\right)\right]\left[Y_{2}\left(s\right)-Y_{2}\left(s'\right)\right]\geqslant0,\text{ for all }s,s'\in S$$

For instance any  $Y \in B(\mathcal{G})$  is comonotonic with any  $c \in \mathbb{R}$ . Moreover, if  $Y_1, Y_2 \in B(\mathcal{G})$ , and if  $Y_2$  is of the form  $Y_2 = I \circ Y_1$ , for some Borel-measurable function I, then  $Y_2$  is comonotonic with  $Y_1$  if and only if the function I is nondecreasing.

**Definition 2.3.** A function  $L: \mathbb{R}^2 \to \mathbb{R}$  is supermodular if for any  $x_1, x_2, y_1, y_2 \in \mathbb{R}$  with  $x_1 \leq x_2$  and  $y_1 \leq y_2$ , one has

$$L(x_2, y_2) + L(x_1, y_1) \ge L(x_1, y_2) + L(x_2, y_1)$$

Appendix B contains more material about supermodular functions on lattices (Definition B.5). For instance, if  $g: \mathbb{R} \to \mathbb{R}$  is concave, and  $a \in \mathbb{R}$ , then the function  $L_1: \mathbb{R}^2 \to \mathbb{R}$  defined by  $L_1(x,y) = g(a-x+y)$  is supermodular (see Example B.7).

2.1. Vigilant Risk Measures and the Weak DC-Property. Let P be a given probability measure on the measurable space  $(S, \mathcal{G})$ . In many situations of choice under uncertainty, the elements of choice are the elements of  $B^+(\mathcal{G})$ , as in the problem that will be examined in this paper. Often, a problem of choice involving these elements is stated as an optimization problem subject to some constraints. In an abstract form, some of these constraints can be stated in terms of operators  $\Psi: B^+(\mathcal{G}) \to \mathbb{R}$ , and might be called "aggregation constraints", or "risk measure constraints". Here I will define two special kinds of these operators (or risk measures): vigilant risk measures, and risk measures having the Weak DC-Property.

**Definition 2.4.** A risk measure is any mapping  $\Psi: B^+(\mathcal{G}) \to \mathbb{R}$ .

**Definition 2.5** (Weak DC-Property). A risk measure  $\Psi: B^+(\mathcal{G}) \to \mathbb{R}$  is said to have the Weak DC-Property if for any  $Y^* \in B^+(\mathcal{G})$  and for any sequence  $\{Y_n\}_{n\geqslant 1} \subset B^+(\mathcal{G})$  such that

- (1)  $\lim_{n\to+\infty} Y_n = Y^*$  (pointwise), and
- (2) there is some  $Z \in B^+(\mathcal{G})$  such that  $Y_n \leq Z$ , for each  $n \geq 1$ ,

the following holds:

$$\lim_{n \to +\infty} \Psi\left(Y_n\right) = \Psi\left(Y^*\right)$$

When  $\Psi$  is defined as a Lebesgue integral with respect to P, i.e.  $\Psi(Y) = \int Y \ dP$  for each  $Y \in B^+(\mathcal{G})$ , then Lebesgue's Dominated Convergence Theorem [13, Th. 2.4.4] implies that  $\Psi$  has the Weak DC-Property. More generally, if  $\Psi$  is a Choquet integral (Appendix A) with respect to some continuous capacity  $\nu$  on  $(S,\mathcal{G})$  (Definition A.2), i.e.  $\Psi(Y) = \int Y \ d\nu$  for each  $Y \in B^+(\mathcal{G})$ , then when seen as an operator on  $B^+(\mathcal{G})$ ,  $\Psi$  has the Weak DC-Property. This is a consequence of [47, Th. 7.16]<sup>6</sup>.

**Definition 2.6.** Recall that P is a probability measure on  $(S, \mathcal{G})$ . A mapping  $\Psi : B(\Sigma) \to \mathbb{R}$  is said to be P-law-invariant, or law-invariant with respect to P, if for any  $\phi_1, \phi_2 \in B(\Sigma)$ ,  $\Psi(\phi_1) = \Psi(\phi_2)$  whenever  $\phi_1$  and  $\phi_2$  have the same distribution according to P.

**Definition 2.7** (Vigilance). Let X be a given element of  $B^+(\mathcal{G})$ , and recall that P is a probability measure on  $(S,\mathcal{G})$ . Denote by  $\Sigma$  the  $\sigma$ -algebra  $\sigma\{X\}$  of subsets of S generated by X. A risk measure  $\Psi: B^+(\Sigma) \to \mathbb{R}$  is said to be (P,X)-vigilant if for any  $Y_1, Y_2 \in B^+(\Sigma)$  such that

- (i)  $Y_1$  and  $Y_2$  have the same distribution under P, i.e.  $P \circ Y_1^{-1} = P \circ Y_2^{-1}$ , and,
- (ii)  $Y_2$  is a nondecreasing function of X, i.e.  $Y_2$  and X are comonotonic,

the following holds:

$$\Psi(Y_2) \leqslant \Psi(Y_1)$$

Clearly, if  $\Psi$  is P-law invariant then it is (P,X)-vigilant. This covers a large collection of operators on  $B^+$  ( $\Sigma$ ) such that a Lebesgue integral with respect to P, a Choquet integral with respect to a distortion of P (Appendix A), and so on. When  $\Psi$  is not P-law invariant, the same intuition as that behind Ghossoub's [23] definition applies here. Namely, the elements of choice are those elements of  $B^+$  ( $\Sigma$ ). Any  $Y \in B^+$  ( $\Sigma$ ) can be written as a function of X, i.e. of the form  $Y = I \circ X$ . One can then think of such Y, rather informally, as possessing two sources of variability: one which results from the state space itself and the indeterminacy of the state of nature at the time of decision, and the other which is associated with the variability of Y with respect to X, i.e. with the variability of the function I with respect to the identity function (on the range of X). The first kind of variability depends on the distribution of X, and it is called baseline randomness by Ghossoub [23]. The second kind of variability does not depend on the distribution of X, and it is called idiosyncratic randomness by Ghossoub [23]. (P,X)-vigilance of the risk measure  $\Psi: B^+$  ( $\Sigma$ )  $\to \mathbb{R}$  can then be understood as requiring that:

- (1) For a given variability of the first kind, fixed according to the probability measure P (that is, when the distribution of X is considered with respect to the measure P), a comparison of two contingent claims through  $\Psi$  can be restricted to a comparison of their idiosyncratic randomness only; and,
- (2) If a given function I's variability is similar to that of the identity function Id (that is, I is comonotonic with Id), then less idiosyncratic randomness is attributed to I than to another function J for which the variability is not similar to that of Id, and hence  $I \circ X$  will be attributed a lower value than  $J \circ X$  through  $\Psi$  (i.e. a lower measure of risk).

<sup>&</sup>lt;sup>6</sup>Theorem 7.16 in Pap [47] is a Dominated Convergence Theorem for the *Šipoš integral*, or the *symmetric Choquet integral*. However, the latter coincides with the Choquet integral for nonnegative functions.

#### 3. Existence of Comonotonic Optimal Contingent Claims

Consider the setting of Section 2, and let X be a given element of  $B^+(\mathcal{G})$  with closed range X(S) = [0, M], where  $M := \|X\|_{sup} < +\infty$ . Denote by  $\Sigma$  the  $\sigma$ -algebra  $\sigma\{X\}$  of subsets of S generated by X, and let P be a probability measure on  $(S, \mathcal{G})$ . I also assume that X is a continuous random variable<sup>7</sup> for P.

Let  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  be a given function, and let  $\rho: B^+(\Sigma) \to \mathbb{R}$  be a given risk measure. The random variable X is fixed, and the objects P,  $\mathcal{U}$ , and  $\rho$  are considered to be the primitives of the following problem:

(3.1) 
$$\sup_{Y \in B(\Sigma)} \left\{ \int \mathcal{U}(X, Y) \ dP \mid 0 \leqslant Y \leqslant X, \ \rho(Y) \leqslant R \right\}$$

where  $R \in \mathbb{R}$  is fixed. The following theorem gives sufficient conditions for the optimal choice  $Y^*$  of the choice variable Y to be a nondecreasing function of X. The proof of the theorem is given in Appendix C.

#### **Theorem 3.1.** If the following hold:

- (1) Problem (3.1) has a nonempty feasibility set,
- (2) The mapping  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  is supermodular.
- (3) The mapping  $\mathcal{E}$  on  $B^+(\Sigma)$  defined by  $\mathcal{E}(Y) := \mathcal{U}(X,Y)$  is uniformly bounded and sequentially continuous in the topology of pointwise convergence.
- (4)  $\rho$  is (P, X)-vigilant, and,
- (5)  $\rho$  has the Weak DC-Property,

then Problem (3.1) admits a solution  $Y^*$  which is comonotonic with X. Moreover, any other solution  $Z^*$  which is comonotonic with X and identically distributed as  $Y^*$  under P is such that  $Y^* = Z^*$ , P-a.s.

A few comments on the assumptions in Theorem 3.1 are in order. First, the assumption of nonemptiness of the feasibility set of Problem (3.1) is made simply to rule out trivial cases where no solution can exist. The assumption of supermodularity of the mapping  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  is not a strong assumption by any means. It is usually given in many situations by the very nature of the problem considered. This happens for instance when  $\mathcal{U}(X,Y) = u(a-X+Y)$ , for a concave utility function u and some  $a \in \mathbb{R}$ . See Example B.7 (1). Assumption (4) in Theorem 3.1 is typically obtained whenever  $\mathcal{U}(X,Y) = u(a-X+Y)$ , for some continuous and bounded utility function u, and some  $a \in \mathbb{R}$ . Assumptions (5) and (6) were discussed in Section 2.1.

Theorem 3.1 is a general result that can be used in many situations. For instance, it can be applied to problems of demand for insurance indemnities as in the model of Arrow [3] and Raviv [48], but also allowing for heterogeneous beliefs unlike the classical approach to that problem.

<sup>&</sup>lt;sup>7</sup>This is a standard assumption (e.g. Carlier and Dana [9, 10], Dana [15], and parts of Schied [52]), and it holds in many instances, such as when it is assumed that a probability density function for X exists.

<sup>&</sup>lt;sup>8</sup>That is, (i) there exists some  $N < +\infty$  such that  $\| \mathcal{E}(Y) \|_{sup} \leq N$  for each  $Y \in B^+(\Sigma)$ ; and, (ii) if  $\{Y_n\}_n$  is a sequence in  $B^+(\Sigma)$  that converges pointwise to some  $Y \in B^+(\Sigma)$ , then the sequence  $\{\mathcal{E}(Y_n)\}_n$  converges pointwise to  $\mathcal{E}(Y)$ .

4. A QUANTILE CHARACTERIZATION OF OPTIMAL CONTINGENT CLAIMS UNDER A DISTORTION RISK MEASURE

Let  $\mathcal{H} := \{Y \in B(\Sigma) \mid 0 \leqslant Y \leqslant X \text{ and } \rho(Y) \leqslant R\}$  denote the feasibility set of Problem (3.1). For each  $Y \in B^+(\Sigma)$ , let  $F_Y(t) := P(\{s \in S : Y(s) \leqslant t\})$  denote the cumulative distribution function (cdf) of Y with respect to the probability measure P, and let  $F_X(t) := P(\{s \in S : X(s) \leqslant t\})$  denote the cdf of X with respect to the probability measure P. Let  $F_Y^{-1}(t)$  be the left-continuous inverse of the cdf  $F_Y$  (i.e. the quantile function of Y), defined by

$$(4.1) F_Y^{-1}(t) := \inf \left\{ z \in \mathbb{R}^+ \mid F_Y(z) \geqslant t \right\}, \ \forall t \in [0, 1]$$

**Lemma 4.1.** If X is a continuous random variable for the probability measure P, then

- (1)  $U := F_X(X)$  is a random variable on the probability space  $(S, \Sigma, P)$  with a uniform distribution on (0,1),
- (2)  $X = F_X^{-1}(U)$ , P-a.s.

If, in addition,  $\rho$  is (P,X)-vigilant, and the mapping  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  is supermodular, then for each  $Y \in \mathcal{H}$ , the function  $Y^*$  defined by  $Y^* := F_Y^{-1}(F_X(X))$  is such that:

- (1)  $Y^* \in \mathcal{H}$ ,
- (2)  $Y^*$  is comonotonic with X,
- (3)  $\int \mathcal{U}(X, Y^*) dP \geqslant \int \mathcal{U}(X, Y) dP$ , and,
- $(4) \ \rho(Y^*) \leqslant \rho(Y).$

The proof of Lemma 4.1 is given in Appendix D. Under the assumptions of Lemma 4.1, it follows that for each  $Y \in \mathcal{H}$  the following holds:

$$(1) \int \mathcal{E}\left(F_Y^{-1}\left(U\right)\right) dP = \int \mathcal{U}\left(F_X^{-1}\left(U\right), F_Y^{-1}\left(U\right)\right) dP \geqslant \int \mathcal{U}\left(X, Y\right) dP$$

$$(2) \rho\left(F_V^{-1}(U)\right) \leqslant \rho(Y).$$

where the mapping  $\mathcal{E}$  on  $B^+(\Sigma)$  is defined by  $\mathcal{E}(Y) = \mathcal{U}(X,Y)$ . Hence, by Lemma 4.1, one can look for a solution to Problem (3.1) of the form  $F^{-1}(U)$ , where F is the cdf of a function  $Z \in B^+(\Sigma)$  such that  $0 \leq Z \leq X$  and  $\rho(Z) \leq R$ .

4.1. **Distortion Risk Measure.** In the following, the risk measure  $\rho$  is assumed to be a distortion risk measure. Specifically, I make the following assumption.

**Assumption 4.2.** There exists a function  $T:[0,1] \to [0,1]$  such that:

- (1) T is increasing and continuously differentiable,
- (2) T(0) = 0 and T(1) = 1,
- (3)  $\rho(Y) = \hat{\int} Y \ dT \circ P$ , for each  $Y \in B^+(\Sigma)$ , where integration is in the sense of Choquet as in Appendix A.

It follows immediately from the definition of vigilance that the distortion risk measure  $\rho$  is (P, X)-vigilant. Moreover, by a classical result [47, Th. 7.16],  $\rho$  has the Weak DC-Property. Hence, Theorem 3.1 yields the existence of a solution to the following problem:

$$(4.2) \qquad \sup_{Y \in B(\Sigma)} \left\{ \int \mathcal{U}(X,Y) \ dP \ \middle| \ 0 \leqslant Y \leqslant X, \ \rho(Y) := \widehat{\int} Y \ dT \circ P \leqslant R \right\}$$

Now, for any given  $Y \in B^{+}(\Sigma)$ , if  $Y^{*} = F_{Y}^{-1}(F_{X}(X)) = F_{Y}^{-1}(U)$ , then one can write

$$\rho\left(Y^{*}\right) = \widehat{\int} F_{Y}^{-1}\left(U\right) \ dT \circ P = \int_{0}^{1} T'\left(1 - t\right) F_{Y}^{-1}\left(t\right) \ dt = \int T'\left(1 - U\right) F_{Y}^{-1}\left(U\right) \ dP$$

and

$$\int \mathcal{U}\left(X,Y^{*}\right) \, dP = \int \mathcal{U}\left(F_{X}^{-1}\left(U\right),F_{Y}^{-1}\left(U\right)\right) = \int \mathcal{E}\left(F_{Y}^{-1}\left(U\right)\right) \, dP = \int_{0}^{1} \mathcal{E}\left(F_{Y}^{-1}\left(t\right)\right) \, dt$$

**Definition 4.3.** Let Q denote the collection of all quantile functions. That is,

$$\mathcal{Q} := \Big\{ f: (0,1) \to \mathbb{R} \ \Big| \ f \text{ is nondecreasing and left-continuous} \Big\}$$

Let  $\mathcal{Q}^*$  denote the collection of all quantile functions  $f \in \mathcal{Q}$  of the form  $F^{-1}$ , where F is the cdf of some function  $Y \in B^+(\Sigma)$  such that  $0 \leq Y \leq X$ . That is,

$$\mathcal{Q}^* = \left\{ f \in \mathcal{Q} \mid 0 \leqslant f(z) \leqslant F_X^{-1}(z), \text{ for each } 0 < z < 1 \right\}$$

4.2. A Quantile Problem. Consider the following problem:

(4.4) 
$$\sup_{f \in \mathcal{Q}^*} \left\{ \int \mathcal{E}\left(f\left(U\right)\right) \ dP \ \middle| \ \int T'\left(1 - U\right) f\left(U\right) \ dP \leqslant R \right\}$$

In light of Lemma 4.1, the following result is immediate.

**Proposition 4.4.** Suppose that the assumptions of Lemma 4.1 hold. If  $f^*$  is optimal for Problem (4.4), then the function  $f^*(U)$  is optimal for Problem (4.2).

Proof. Suppose that  $f^* \in \mathcal{Q}^*$  is optimal for Problem (4.4), and let  $Z^* \in B^+(\Sigma)$  be a corresponding function. That is,  $f^*$  is the quantile function of  $Z^*$ . Hence,  $0 \leq Z^* \leq X$ , and  $\rho(Z^*) = \int Z^* dT \circ P = \int T'(1-U) F_{Z^*}^{-1}(U) dP = \int T'(1-U) f^*(U) dP \leq R$ . Therefore,  $Z^*$  is feasible for Problem (4.2). Let  $\tilde{Z}^* := f^*(U)$ . Then, by Lemma 4.1,  $\tilde{Z}^* = f^*(U)$  is feasible for Problem (4.2). To show optimality, let Z be any feasible solution for Problem (4.2), and let F be the cdf of Z. Then, by Lemma 4.1, the function  $\tilde{Z} := F^{-1}(U)$  is feasible for Probem (4.2), comonotonic with X, and satisfies:

• 
$$\int \mathcal{U}(X, \widetilde{Z}) dP \geqslant \int \mathcal{U}(X, Z) dP$$
, and,

<sup>&</sup>lt;sup>9</sup>This is a standard exercise. See, for instance, [34, p. 418] or [29, p. 213].

• 
$$\widehat{\int} \widetilde{Z} dT \circ P \leqslant \widehat{\int} Z dT \circ P \leqslant R$$
.

Moreover,  $\widetilde{Z}$  has also F as a cdf. To show optimality of  $\widetilde{Z}^* = f^*(U)$  for Problem (4.2) it remains to show that

$$\int \mathcal{U}\left(X, \widetilde{Z}^*\right) dP \geqslant \int \mathcal{U}\left(X, \widetilde{Z}\right) dP$$

Now, let  $f := F^{-1}$ , so that  $\widetilde{Z} = f(U)$ . Since  $\widetilde{Z}$  is feasible for Probem (4.2), it follows that

$$R \geqslant \rho\left(\widetilde{Z}\right) = \int T'(1-U) f(U) dP$$

Hence, f is feasible for Problem (4.4). Since  $f^*$  is optimal for Problem (4.4), it follows that

$$\int \mathcal{E}\left(f^{*}\left(U\right)\right) \ dP \geqslant \int \mathcal{E}\left(f\left(U\right)\right) \ dP$$

That is

$$\int \mathcal{U}\left(X,\widetilde{Z}^{*}\right) \, dP \geqslant \int \mathcal{U}\left(X,\widetilde{Z}\right) \, dP$$

Therefore,  $\widetilde{Z}^* = f^*(U)$  is optimal for Problem (4.2).

5. An Example: Utility Maximization with a Convex Distortion Risk Measure Consider the previous setting, and assume the following.

**Assumption 5.1.** There exists a utility function  $u : \mathbb{R} \to \mathbb{R}$  such that:

- (1) u is bounded and satisfies Inada's [32] conditions. That is:
  - u is bounded,
  - u(0) = 0,
  - u is strictly increasing and strictly concave,
  - u is continuously differentiable, and,
  - $u'(0) = +\infty$  and  $\lim_{x \to +\infty} u'(x) = 0$ .
- (2) For each  $Y \in B^+(\Sigma)$ ,  $\mathcal{E}(Y) = \mathcal{U}(X,Y) = u(W_0 + Y X)$ .

The strict concavity and the continuous differentiability of u imply that u' is both continuous and strictly decreasing. The latter implies that  $(u')^{-1}$  is continuous and strictly decreasing, by the Inverse Function Theorem [50, pp. 221-223].

**Assumption 5.2.** The DM has initial wealth  $W_0$  such that  $X \leq W_0$ , P-a.s. That is,

$$P\bigg(\Big\{s \in S : X\left(s\right) > W_0\Big\}\bigg) = 0$$

Assumption 5.2 simply states that the DM is well-diversified so that the particular exposure to X is sufficiently small, with respect to the DM's total portfolio exposure.

5.1. **A Utility Maximization Problem.** The initial Problem (3.1) then becomes the following utility maximization problem.

$$(5.1) \qquad \sup_{Y \in B(\Sigma)} \left\{ \int u \left( W_0 + Y - X \right) \ dP \ \middle| \ 0 \leqslant Y \leqslant X, \ \rho \left( Y \right) := \widehat{\int} Y \ dT \circ P \leqslant R \right\}$$

The concavity of the utility function u (Assumption 5.1) implies that the function  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  defined by  $\mathcal{U}(x,y) = u(y-x)$  is supermodular (Example B.7 (1)). Moreover, the risk measure  $\rho$  given by  $\rho(Y) = \widehat{\int} Y \ dT \circ P$  is (P,X)-vigilant and has the Weak DC-Property. Hence, Theorem 3.1 yields the existence of a solution to Problem (5.1).

**Remark 5.3.** For any  $Y \in B(\Sigma)$  which is feasible for Problem 5.1, one has  $0 \le Y \le X$ . Therefore, by monotonicity of the Choquet integral (Proposition A.4 (5)), it follows that

$$0 \leqslant \widehat{\int} Y \ dT \circ P \leqslant \widehat{\int} X \ dT \circ P$$

Hence, for the risk measure constraint  $\rho(Y) \leq R$  to be meaningful in the present context, it will be assumed that  $R \leq \widehat{\int} X \ dT \circ P$ . This will then imply that any  $Y \in B(\Sigma)$  which is feasible for Problem 5.1 satisfies:

$$0 \leqslant \widehat{\int} Y \ dT \circ P \leqslant R \leqslant \widehat{\int} X \ dT \circ P$$

Now, let  $U = F_X(X)$ , and consider the following problem.

$$(5.2) \qquad \sup_{f \in \mathcal{Q}^*} \left\{ \int u \left( W_0 + f \left( U \right) - F_X^{-1} \left( U \right) \right) dP \, \middle| \, \int T' \left( 1 - U \right) f \left( U \right) dP \leqslant R \right\}$$

where  $Q^*$  is given by Equation (4.3).

**Proposition 5.4.** Suppose that the assumptions of Lemma 4.1 hold. If  $f^*$  is optimal for Problem (5.2), then the function  $f^*(U)$  is optimal for Problem (5.1).

*Proof.* By Lemma 4.1,  $X = F_X^{-1}(U)$ , P-a.s. Therefore, for any quantile function  $f \in \mathcal{Q}$ ,

$$\int u\left(W_{0}+f\left(U\right)-F_{X}^{-1}\left(U\right)\right) \ dP = \int u\left(W_{0}+f\left(U\right)-X\right) \ dP = \int \mathcal{E}\left(f\left(U\right)\right) \ dP$$

The rest follows from Proposition 4.4.

Note that Proposition 5.4 holds for any distortion function T, and the convexity of T has not been assumed yet.

### 5.2. Characterizing a Solution.

**Lemma 5.5.** Suppose that X is a continuous random variable for the probability measure P. If  $f^* \in \mathcal{Q}^*$  satisfies the following:

- (1)  $\int_0^1 T'(1-t) f^*(t) dt = R$ ,
- (2) There exists  $\lambda \ge 0$  such that for all  $t \in (0,1)$ ,

$$f^{*}(t) = \underset{0 \leq y \leq F_{X}^{-1}(t)}{\arg \max} \left[ u \left( W_{0} + y - F_{X}^{-1}(t) \right) - \lambda T'(1 - t) y \right],$$

then  $f^*$  solves Problem (5.2).

If, in addition, the rest of the assumptions of Lemma 4.1 hold, then the function  $f^*(U)$  is optimal for Problem (5.1), where  $U = F_X(X)$ .

*Proof.* Suppose that  $f^* \in \mathcal{Q}^*$  satisfies conditions (1) and (2) above. Then, in particular,  $f^*$  is feasible for Problem (5.2). To show optimality of  $f^*$  for Problem (5.2), let f by any other feasible solution for Problem (5.2). Then  $\int_0^1 T'(1-t) f(t) dt \leq R$  and, for all  $t \in (0,1)$ ,

$$u\left(W_{0} + f^{*}(t) - F_{X}^{-1}(t)\right) - \lambda T'(1-t) f^{*}(t) \ge u\left(W_{0} + f(t) - F_{X}^{-1}(t)\right) - \lambda T'(1-t) f(t),$$
that is,

$$\left[u\left(W_{0}+f^{*}\left(t\right)-F_{X}^{-1}\left(t\right)\right)-u\left(W_{0}+f\left(t\right)-F_{X}^{-1}\left(t\right)\right)\right]\geqslant\lambda T'\left(1-t\right)\left[f^{*}\left(t\right)-f\left(t\right)\right]$$

Integrating yields

$$\int_{0}^{1} u \Big( W_{0} + f^{*}(t) - F_{X}^{-1}(t) \Big) dt - \int_{0}^{1} u \Big( W_{0} + f(t) - F_{X}^{-1}(t) \Big) dt$$

$$\geqslant \lambda \left[ R - \int_{0}^{1} T'(1 - t) f(t) dt \right] \geqslant 0$$

or,

$$\int u \Big( W_0 + f^* (U) - F_X^{-1} (U) \Big) dP = \int_0^1 u \Big( W_0 + f^* (t) - F_X^{-1} (t) \Big) dt$$

$$\geqslant \int_0^1 u \Big( W_0 + f (t) - F_X^{-1} (t) \Big) dt = \int u \Big( W_0 + f (U) - F_X^{-1} (U) \Big) dP$$

as required. The rest follows from Proposition 5.4.

Lemma 5.5 suggests that in order to find a solution for Problem (5.2), one can start by solving the problem

(5.3) 
$$\max_{0 \leqslant f_{\lambda}(t) \leqslant F_{X}^{-1}(t)} \left[ u \left( W_{0} + f_{\lambda}(t) - F_{X}^{-1}(t) \right) - \lambda T'(1-t) f_{\lambda}(t) \right]$$

for a given  $\lambda \ge 0$ , and for a fixed  $t \in (0,1)$ .

**Assumption 5.6.** T is convex and  $T'(1) < +\infty$ .

Assumption 5.6 yields an explicit characterization of an optimal solution to Problem (5.1), as given in the following Corollary.

**Corollary 5.7.** Suppose that the assumptions of Lemma 4.1 hold, and let  $U = F_X(X)$ . If Assumption 4.2, Assumption 5.1, Assumption 5.2, and Assumption 5.6 also hold, then an optimal solution for Problem (5.1) takes the form:

(5.4) 
$$\mathcal{Y}^* = \max \left[ 0, \min \left\{ F_X^{-1}(U), (u')^{-1} \left( \lambda^* T'(1-U) \right) + F_X^{-1}(U) - W_0 \right\} \right]$$

where  $\lambda^*$  is chosen so that  $\int T'(1-U)\mathcal{Y}^* dP = R$ .

*Proof.* For a given  $\lambda \geq 0$ , and for a fixed  $t \in (0,1)$ , consider the problem:

(5.5) 
$$\max_{f_{\lambda}(t)} \left[ u \left( W_0 + f_{\lambda}(t) - F_X^{-1}(t) \right) - \lambda T'(1-t) f_{\lambda}(t) \right]$$

By Assumption 5.1, the first-order conditions are sufficient for an optimum for Problem (5.5), and they imply that the function

$$f_{\lambda}^{*}(t) := (u')^{-1} (\lambda T'(1-t)) + F_{X}^{-1}(t) - W_{0}$$

solves Problem (5.5). Concavity of u and convexity of T imply that the function  $f_{\lambda}^*:(0,1)\to\mathbb{R}$  is nondecreasing, since  $F_X^{-1}$  is a nondecreasing function. Assumption 4.2 and Assumption 5.1 also yield left-continuity of  $f_{\lambda}^*$ . Consequently,  $f_{\lambda}^*\in\mathcal{Q}$ , the set of all quantile functions.

Now, define the function  $f_{\lambda}^{**}$  by

(5.6) 
$$f_{\lambda}^{**}\left(t\right) := \max\left[0, \min\left\{F_{X}^{-1}\left(t\right), f_{\lambda}^{*}\left(t\right)\right\}\right]$$

It is then easy to check that  $f_{\lambda}^{**} \in \mathcal{Q}$ , since  $f_{\lambda}^{*} \in \mathcal{Q}$  and since  $F_{X}^{-1}$  is a nondecreasing function. Moreover,  $0 \leq f_{\lambda}^{**}(z) \leq F_{X}^{-1}(z)$ , for each  $z \in (0,1)$ . Therefore,  $f_{\lambda}^{**} \in \mathcal{Q}^{*}$ . Finally, it is easily seen that  $f_{\lambda}^{**}(t)$  solves Problem (5.3) for the given  $\lambda$  and t, since the concavity of u yields the concavity of the function  $z \mapsto u\Big(W_0 + z - F_X^{-1}(t)\Big) - \lambda T'(1-t)z$ , for each  $t \in (0,1)$ . Hence, in view of Lemma 5.5, it remains to show that there exists a  $\lambda^* \geq 0$  such that  $\int_0^1 T'(1-t) f_{\lambda}^{**}(t) dt = R$ . This is given by Lemma E.1 in Appendix E.

#### 6. Conclusion

In this paper I examined a general utility maximization problem, with a non-law-invariant indirect utility and a non-law-invariant risk measure constraint. This problem of demand for contingent claims is an abstraction of many common problems in economic theory that were hitherto only considered in a framework of complete homogeneity of beliefs about the realizations of an underlying uncertainty. The problem examined in this paper takes the following form:

(6.1) 
$$\sup_{Y \in B(\Sigma)} \left\{ \int \mathcal{U}(X,Y) \ dP \ \middle| \ 0 \leqslant Y \leqslant X, \ \rho(Y) \leqslant R \right\}$$

where X is a given underlying uncertainty on some state space S,  $\Sigma = \sigma\{X\}$ ,  $Y = I \circ X$  is a claim contingent on this uncertainty,  $\int \mathcal{U}(X, I \circ X) \ dP$  is a DM's expected utility of wealth with respect to the probability measure P,  $\rho : B^+(\Sigma) \to \mathbb{R}$  is a given risk measure, and  $R \in \mathbb{R}$  is fixed. The set  $B(\Sigma)$  is the linear space of all bounded and  $\Sigma$ -measurable functions on S.

Conditions on the primitives  $\mathcal{U}$  and  $\rho$  for the optimal choice  $Y^*$  of Y to be monotone in X are desired. Using some quantile reformulation techniques inspired by the theory of monotone equimeasurable rearrangements, I showed that a set of sufficient conditions on  $\mathcal{U}$  and  $\rho$  for this to hold is that the risk measure  $\rho$  be Vigilant and satisfy the Weak DC-Property, and that the function  $\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  be supermodular. Roughly speaking, vigilance of the risk measure  $\rho$  can be understood as a (weak) preference for comonotonicity, on the collection of all functions that are identically distributed for the probability measure P. The Weak DC-Property is verified by a large class of operators on  $B^+(\Sigma)$ , such as the Lebesgue integral or the Choquet integral (Appendix A). For instance, any distortion risk measure of the form  $\rho(Y) = \hat{\int} Y \ dP$  is (P, X)-vigilant. In this case, I gave a characterization of an optimal contingent claim in terms of a quantile problem, in which the uncertainty may be seen as being homogeneous. In the special case of a convex distortion risk measure, a full explicit characterization of an optimal contingent claim was given.

#### APPENDIX A. RELATED ANALYSIS

#### A.1. A Useful Result.

**Lemma A.1.** If  $(f_n)_n$  is a uniformly bounded sequence of nondecreasing real-valued functions on some closed interval  $\mathcal{I}$  in  $\mathbb{R}$ , with bound N (i.e.  $|f_n(x)| \leq N$ ,  $\forall x \in \mathcal{I}$ ,  $\forall n \geq 1$ ), then there exists a nondecreasing real-valued bounded function  $f^*$  on  $\mathcal{I}$ , also with bound N, and a subsequence of  $(f_n)_n$  that converges pointwise to  $f^*$  on  $\mathcal{I}$ .

Proof. [11, Lemma 13.15]. 
$$\Box$$

#### A.2. Capacities and the Choquet Integral.

**Definition A.2.** A (normalized) *capacity* on a measurable space  $(S, \Sigma)$  is a set function  $\nu : \Sigma \to [0, 1]$  such that

- (1)  $\nu(\emptyset) = 0$ :
- (2)  $\nu(S) = 1$ ; and,
- (3)  $\nu$  is monotone: for any  $A, B \in \Sigma$ ,  $A \subseteq B \Rightarrow \nu(A) \leqslant \nu(B)$ .

The capacity  $\nu$  is said to be

- (1) Continuous from above if for any sequence  $\{A_n\}_n$  in  $\mathcal{G}$  such that  $A_{n+1} \subseteq A_n$  for each  $n \ge 1$ , one has  $\lim_{n \to +\infty} \nu(A_n) = \nu(\bigcap_{n=1}^{+\infty} A_n)$ .
- (2) Continuous from below if for any sequence  $\{A_n\}_n$  in  $\mathcal{G}$  such that  $A_n \subseteq A_{n+1}$  for each  $n \ge 1$ , one has  $\lim_{n \to +\infty} \nu(A_n) = \nu(\bigcup_{n=1}^{+\infty} A_n)$ .
- (3) Continuous if it is both continuous from above and continuous from below.

For instance, if P is a probability measure on  $(S, \Sigma)$  and  $T : [0, 1] \to [0, 1]$  is increasing, with T(0) = 0 and T(1) = 1, then the set function  $\nu := T \circ P$  is a capacity on  $(S, \Sigma)$ . Such a function T is usually called a *probability distortion*, and the capacity  $T \circ P$  is usually called a *distorted probability measure*. If, moreover, the function T is continuous, then the set function  $\nu := T \circ P$  is a capacity on  $(S, \Sigma)$  which is continuous. This is an immediate consequence of the continuity of the measure P for monotone sequences [13, Prop. 1.2.3] and the continuity of T. In particular, any probability measure is continuous.

**Definition A.3.** For a given capacity  $\nu$  on  $(S, \Sigma)$  and a given  $\phi \in B(\Sigma)$ , the *Choquet integral* of  $\phi$  with respect to  $\nu$  is defined by

$$\widehat{\int} \phi \ d\nu := \int_0^{+\infty} \nu \left( \left\{ s \in S : \phi \left( s \right) \geqslant t \right\} \right) \ dt + \int_{-\infty}^0 \left[ \nu \left( \left\{ s \in S : \phi \left( s \right) \geqslant t \right\} \right) - 1 \right] \ dt$$

where the integrals are taken in the sense of Riemann.

The Choquet integral with respect to a measure is simply the usual Lebesgue integral with respect to that measure [39, p. 59]. Unlike the Lebesgue integral, however, the Choquet integral is not an additive operator on  $B(\Sigma)$ . However, the Choquet integral is additive on comonotonic functions (Definition 2.2).

**Proposition A.4.** Let  $\nu$  be a capacity on  $(S, \Sigma)$ .

- (1) If  $\phi_1, \phi_2 \in B(\Sigma)$  are comonotonic, then  $\hat{\int} (\phi_1 + \phi_2) d\nu = \hat{\int} \phi_1 d\nu + \hat{\int} \phi_2 d\nu$ .
- (2) If  $\phi \in B(\Sigma)$  and  $c \in \mathbb{R}$ , then  $\widehat{\int} (\phi + c) d\nu = \widehat{\int} \phi d\nu + c$ .
- (3) If  $A \in \Sigma$  then  $\widehat{\int} \mathbf{1}_A d\nu = \nu(A)$ .
- (4) If  $\phi \in B(\Sigma)$  and  $a \ge 0$ , then  $\widehat{\int} a \phi d\nu = a \widehat{\int} \phi d\nu$ .
- (5) If  $\phi_1, \phi_2 \in B(\Sigma)$  are such that  $\phi_1 \leqslant \phi_2$ , then  $\widehat{\int} \phi_1 \ d\nu \leqslant \widehat{\int} \phi_2 \ d\nu$ .

For more about capacities and Choquet integrals, I refer to Marinacci and Montrucchio [39].

#### APPENDIX B. EQUIMEASURABLE MONOTONE REARRANGEMENTS AND SUPERMODULARITY

The concept of an equimeasurable rearrangement of a Borel-measurable function on  $\mathbb{R}$  with respect to a finite Borel measure, and the notion of an equimeasurable rearrangement of a measurable function f from a measurable space into  $\mathbb{R}$  with respect to a finite Borel measure on the range of f is by now part of the classical literature [7, 12, 28, 38]. It is the basic tool that will be used to show the existence of a monotone solution to the DM's problem. Here, the idea of an equimeasurable rearrangement of a random variable with respect to another random variable is discussed. The nomenclature used has been chosen with the present context in mind, whereby the same measurable space may be endowed with different measures. In this section I introduce a specific formulation of the nondecreasing rearrangement of any element Y of  $B^+(\Sigma)$  with respect to the fixed underlying uncertainty X. Although some of the results presented here are not new, the approach is novel, to the best of my knowledge. All proofs and some additional results may be found in Ghossoub [26] or Ghossoub [24]. The latter introduces the idea of a rearrangement in a context of non-additive probability measures (capacities).

B.1. The Nondecreasing Rearrangement. Let  $(S, \mathcal{G}, P)$  be a probability space, and let  $X \in B^+(\mathcal{G})$  be a continuous random variable (i.e.  $P \circ X^{-1}$  is nonatomic) with range [0, M] := X(S), where  $M := \sup\{X(s) : s \in S\} < +\infty$ , i.e. X is a mapping of S onto the closed interval [0, M]. Denote by  $\Sigma$  the  $\sigma$ -algebra  $\sigma\{X\}$ , and denote by  $\phi$  the law of X defined by

$$\phi(B) := P\left(\left\{s \in S : X(s) \in B\right\}\right) = P \circ X^{-1}(B)$$

for any Borel subset B of  $\mathbb{R}$ .

If  $I, I_n : [0, M] \to [0, M]$ , for each  $n \ge 1$ , I will write  $I_n \downarrow I$ ,  $\phi$ -a.s., to signify that the sequence  $\{I_n\}_n$  is a nonincreasing sequence of functions and that  $\lim_{n \to +\infty} I_n(t) = I(t)$ , for  $\phi$ -a.a.  $t \in [0, M]$ . Similarly, I will write  $I_n \uparrow I$ ,  $\phi$ -a.s., to signify that the sequence  $\{I_n\}_n$  is a nondecreasing sequence of functions and that  $\lim_{n \to +\infty} I_n(t) = I(t)$ , for  $\phi$ -a.a.  $t \in [0, M]$ .

**Definition B.1.** For any Borel-measurable map  $I : [0, M] \to \mathbb{R}$ , define the distribution function of I as the map  $\phi_I : \mathbb{R} \to [0, 1]$  defined by

(B.1) 
$$\phi_I(t) := \phi\left(\left\{x \in [0, M] : I(x) \leqslant t\right\}\right)$$

Then  $\phi_I$  is a nondecreasing right-continuous function, and the function  $t \mapsto 1 - \phi_I(t)$  is called the *survival function* of I.

**Definition B.2.** Let  $I:[0,M] \to [0,M]$  be any Borel-measurable map, and define the function  $\widetilde{I}:[0,M] \to \mathbb{R}$  by

(B.2) 
$$\widetilde{I}(t) := \inf \left\{ z \in \mathbb{R}^+ \mid \phi_I(z) \geqslant \phi([0, t]) \right\}$$

The following proposition gives some useful properties of the map  $\widetilde{I}$  defined above.

**Proposition B.3.** Let  $I:[0,M] \to [0,M]$  be any Borel-measurable map and let  $\widetilde{I}:[0,M] \to \mathbb{R}$  be defined as in equation (B.2). Then the following hold:

- $(1) \ \widetilde{I} \ is \ left-continuous, \ nondecreasing, \ and \ Borel-measurable;$
- (2) For each  $t \in [0, M]$ ,  $\phi_I(\widetilde{I}(t)) \geqslant \phi([0, t])$ ;
- (3)  $\widetilde{I}(t) \ge 0$ , for each  $t \in [0, M]$ ,  $\widetilde{I}(0) = 0$ , and  $\widetilde{I}(M) \le M$ ;
- (4) If  $I_1, I_2 : [0, M] \rightarrow [0, M]$  are such that  $I_1 \leqslant I_2$ ,  $\phi$ -a.s., then  $\widetilde{I}_1 \leqslant \widetilde{I}_2$ ;
- (5) If  $Id: [0, M] \rightarrow [0, M]$  denotes the identity function, then  $\widetilde{I}d \leq Id$ ;
- (6)  $\widetilde{I}$  is  $\phi$ -equimeasurable with I, in the sense that for any Borel set B,

$$\phi\Big(\Big\{t\in[0,M]:I\left(t\right)\in B\Big\}\Big)=\phi\Big(\Big\{t\in[0,M]:\widetilde{I}\left(t\right)\in B\Big\}\Big)$$

(7) If  $\overline{I}:[0,M] \to \mathbb{R}^+$  is another nondecreasing, Borel-measurable map which is  $\phi$ -equimeasurable with I, then  $\overline{I}=\widetilde{I}, \ \phi$ -a.s.;

(8) If 
$$I, I_n : [0, M] \to [0, M]$$
, for each  $n \ge 1$ , and  $I_n \downarrow I$ ,  $\phi$ -a.s., then  $\widetilde{I}_n \downarrow \widetilde{I}$ ,  $\phi$ -a.s.

 $\widetilde{I}$  will be called the nondecreasing  $\phi$ -rearrangement of I (see also [20, pp. 224-225]). Now, define  $Y:=I\circ X$  and  $\widetilde{Y}:=\widetilde{I}\circ X$ . Since both I and  $\widetilde{I}$  are Borel-measurable mappings of [0,M] into itself, it follows that  $Y,\widetilde{Y}\in B^+(\Sigma)$ . Note also that  $\widetilde{Y}$  is nondecreasing in X, in the sense that if  $s_1,s_2\in S$  are such that  $X(s_1)\leqslant X(s_2)$  then  $\widetilde{Y}(s_1)\leqslant \widetilde{Y}(s_2)$ , and that Y and  $\widetilde{Y}$  are P-equimeasurable, that is, for any  $\alpha\in[0,M]$ ,  $P\left(\left\{s\in S:Y(s)\leqslant\alpha\right\}\right)=P\left(\left\{s\in S:\widetilde{Y}(s)\leqslant\alpha\right\}\right)$ .

Call  $\widetilde{Y}$  a nondecreasing P-rearrangement of Y with respect to X, and denote it by  $\widetilde{Y}_P$  to avoid confusion in case a different measure on  $(S,\mathcal{G})$  is also considered. For example, in case both  $P_1$  and  $P_2$  are probability measures on the measurable space  $(S,\mathcal{G})$ , denote by  $\widetilde{Y}_{P_1}$  (resp.  $\widetilde{Y}_{P_2}$ ) a nondecreasing  $P_1$ -rearrangement (resp.  $P_2$ -rearrangement) of Y with respect to X. In the general case, nothing can be said a priori about the relationship between  $\widetilde{Y}_{P_1}$  and  $\widetilde{Y}_{P_2}$ . What can be asserted, however, is that:

- (1) Both  $\widetilde{Y}_{P_{1}}$  and  $\widetilde{Y}_{P_{2}}$  are nondecreasing in X, and hence  $\widetilde{Y}_{P_{1}}$  and  $\widetilde{Y}_{P_{2}}$  are comonotonic, i.e.  $\left[\widetilde{Y}_{P_{2}}\left(s\right)-\widetilde{Y}_{P_{2}}\left(s'\right)\right]\left[\widetilde{Y}_{P_{1}}\left(s\right)-\widetilde{Y}_{P_{1}}\left(s'\right)\right]\geqslant0$ , for all  $s,s'\in S$ ;
- (2) Y and  $\widetilde{Y}_{P_1}$  are  $P_1$ -equimeasurable; and,
- (3) Y and  $\widetilde{Y}_{P_2}$  are  $P_2$ -equimeasurable.

Note that  $\widetilde{Y}_P$  is P-a.s. unique. Note also that if  $Y_1$  and  $Y_2$  are P-equimeasurable and if  $Y_1 \in L_1(S, \mathcal{G}, P)$ , then  $Y_2 \in L_1(S, \mathcal{G}, P)$  and  $\int \psi(Y_1) \ dP = \int \psi(Y_2) \ dP$ , for any measurable function  $\psi$  such that the integrals exist.

Similarly to the previous construction, for a given a Borel-measurable  $B \subseteq [0, M]$  with  $\phi(B) > 0$ , there exists a  $\phi$ -a.s. unique (on B) nondecreasing, Borel-measurable mapping  $\widetilde{I}_B : B \to [0, M]$  which is  $\phi$ -equimeasurable with I on B, in the sense that for any  $\alpha \in [0, M]$ ,

$$\phi\left(\left\{t \in B : I\left(t\right) \leqslant \alpha\right\}\right) = \phi\left(\left\{t \in B : \widetilde{I}_{B}\left(t\right) \leqslant \alpha\right\}\right)$$

 $\widetilde{I}_B$  is called the nondecreasing  $\phi$ -rearrangement of I on B. Since X is  $\mathcal{G}$ -measurable, there exists  $A \in \mathcal{G}$  such that  $A = X^{-1}(B)$ , and hence P(A) > 0. Now, define  $\widetilde{Y}_A := \widetilde{I}_B \circ X$ . Since both I and  $\widetilde{I}_B$  are bounded Borel-measurable mappings, it follows that  $Y, \widetilde{Y}_A \in B^+(\Sigma)$ . Note also that  $\widetilde{Y}_A$  is nondecreasing in X on A, in the sense that if  $s_1, s_2 \in A$  are such that  $X(s_1) \leq X(s_2)$  then  $\widetilde{Y}(s_1) \leq \widetilde{Y}(s_2)$ , and that Y and  $\widetilde{Y}_A$  are P-equimeasurable on A, that is, for any  $\alpha \in [0, M]$ ,  $P\left(\left\{s \in S : Y(s) \leq \alpha\right\} \cap A\right) = P\left(\left\{s \in S : \widetilde{Y}_A(s) \leq \alpha\right\} \cap A\right)$ .

Call  $Y_A$  a nondecreasing P-rearrangement of Y with respect to X on A, and denote it by  $\widetilde{Y}_{A,P}$  to avoid confusion in case a different measure on  $(S,\mathcal{G})$  is also considered. For example, in case both  $P_1$  and  $P_2$  are probability measures on the measurable space  $(S,\mathcal{G})$ , denote by  $\widetilde{Y}_{A,P_1}$  (resp.  $\widetilde{Y}_{A,P_2}$ ) a nondecreasing  $P_1$ -rearrangement (resp.  $P_2$ -rearrangement) of Y with respect to X on A. In the general case, nothing can be said a priori about the relationship between  $\widetilde{Y}_{A,P_1}$  and  $\widetilde{Y}_{A,P_2}$ . What can be asserted, however, is that:

- (1) Both  $\widetilde{Y}_{A,P_1}$  and  $\widetilde{Y}_{A,P_2}$  are nondecreasing in X on A, and hence  $\widetilde{Y}_{P_1}$  and  $\widetilde{Y}_{P_2}$  are comonotonic on A, i.e.  $\left[\widetilde{Y}_{A,P_2}\left(s\right)-\widetilde{Y}_{A,P_2}\left(s'\right)\right]\left[\widetilde{Y}_{A,P_1}\left(s\right)-\widetilde{Y}_{A,P_1}\left(s'\right)\right]\geqslant 0$ , for all  $s,s'\in A$ ;
- (2) Y and  $\widetilde{Y}_{A,P_1}$  are  $P_1$ -equimeasurable on A; and,
- (3) Y and  $\widetilde{Y}_{A,P_2}$  are  $P_2$ -equimeasurable on A.

Note that  $\widetilde{Y}_{A,P}$  is P-a.s. unique. Note also that if  $Y_{1,A}$  and  $Y_{2,A}$  are P-equimeasurable on A and if  $\int_A Y_{1,A} \ dP < +\infty$ , then  $\int_A Y_{2,A} \ dP < +\infty$  and  $\int_A \psi \left( Y_{1,A} \right) \ dP = \int_A \psi \left( Y_{2,A} \right) \ dP$ , for any measurable function  $\psi$  such that the integrals exist.

**Lemma B.4.** Let  $Y \in B^+(\Sigma)$  and let  $A \in \mathcal{G}$  be such that P(A) = 1 and X(A) is a Borel set  $\widetilde{Y}_P$  be the nondecreasing P-rearrangement of Y with respect to X, and let  $\widetilde{Y}_{A,P}$  be the nondecreasing P-rearrangement of Y with respect to X on A. Then  $\widetilde{Y}_P = \widetilde{Y}_{A,P}$ , P-a.s.

B.2. Supermodularity and Hardy-Littlewood-Pólya Inequalities. A partially ordered set (poset) is a pair  $(T, \geq)$  where  $\geq$  is a reflexive, transitive and antisymmetric binary relation on T. A point  $t \in T$  is called an upper bound (resp. lower bound) for a subset S of T if  $t \geq x$  (resp.  $x \geq t$ ) for each  $x \in S$ . A point  $t^* \in T$  is called a least upper bound (resp. greatest lower bound) for S if it is an upper bound (resp. lower bound) for S and for any other upper bound (resp. lower bound) S one has S one ha

For instance, the Euclidian space  $\mathbb{R}^n$  is a lattice for the partial order  $\geq$  defined as follows: for  $x = (x_1, \dots, x_n) \in \mathbb{R}^n$  and  $y = (y_1, \dots, y_n) \in \mathbb{R}^n$ , write  $x \geq y$  when  $x_i \geq y_i$ , for each  $i = 1, \dots, n$ . It is then easy to see that  $x \vee y = (\max(x_1, y_1), \dots, \max(x_n, y_n))$  and  $x \wedge y = (\min(x_1, y_1), \dots, \min(x_n, y_n))$ .

**Definition B.5.** Let  $(T, \geq)$  be a lattice. A function  $L: T \to \mathbb{R}$  is said to be *supermodular* if for each  $x, y \in T$ ,

(B.4) 
$$L(x \lor y) + L(x \land y) \geqslant L(x) + L(y)$$

In particular, a function  $L: \mathbb{R}^2 \to \mathbb{R}$  is supermodular if for any  $x_1, x_2, y_1, y_2 \in \mathbb{R}$  with  $x_1 \leq x_2$  and  $y_1 \leq y_2$ , one has

(B.5) 
$$L(x_2, y_2) + L(x_1, y_1) \ge L(x_1, y_2) + L(x_2, y_1)$$

A function  $L: \mathbb{R}^2 \to \mathbb{R}$  is called *strictly supermodular* if for any  $x_1, x_2, y_1, y_2 \in \mathbb{R}$  with  $x_1 < x_2$  and  $y_1 < y_2$ , one has

(B.6) 
$$L(x_2, y_2) + L(x_1, y_1) > L(x_1, y_2) + L(x_2, y_1)$$

<sup>&</sup>lt;sup>10</sup>Note that if  $A \in \Sigma = \sigma\{X\}$  then X(A) is automatically a Borel set, by definition of  $\sigma\{X\}$ . Indeed, for any  $A \in \sigma\{X\}$ , there is some Borel set B such that  $A = X^{-1}(B)$ . Then  $X(A) = B \cap X(S)$  [17, p. 7]. Thus  $X(A) = B \cap [0, M]$  is a Borel subset of [0, M].

**Lemma B.6.** A function  $L: \mathbb{R}^2 \to \mathbb{R}$  is supermodular (resp. strictly supermodular) if and only if the function  $\eta(y) := L(x+h,y) - L(x,y)$  is nondecreasing (resp. increasing) on  $\mathbb{R}$ , for any  $x \in \mathbb{R}$  and  $h \ge 0$  (resp. h > 0).

**Example B.7.** The following are useful examples of supermodular functions:

- (1) If  $g : \mathbb{R} \to \mathbb{R}$  is concave, and  $a \in \mathbb{R}$ , then the function  $L_1 : \mathbb{R}^2 \to \mathbb{R}$  defined by  $L_1(x,y) = g(a-x+y)$  is supermodular. If, moreover, g is strictly concave then  $L_1$  is strictly supermodular.
- (2) If  $f: \mathbb{R} \to \mathbb{R}$  is concave, and  $a \in \mathbb{R}$ , then the function  $L_2: \mathbb{R}^2 \to \mathbb{R}$  defined by  $L_2(x,y) = f(a+x-y)$  is supermodular. If, moreover, f is strictly concave then  $L_2$  is strictly supermodular.
- (3) The function  $L_3: \mathbb{R}^2 \to \mathbb{R}$  defined by  $L_3(x,y) = -(y-x)^+$  is supermodular.
- (4) If  $\psi, \phi : \mathbb{R} \to \mathbb{R}$  are both nonincreasing or both nondecreasing functions, then the function  $L_4 : \mathbb{R}^2 \to \mathbb{R}$  defined by  $L_4(x,y) = \phi(x) \psi(y)$  is supermodular.

**Lemma B.8** (Hardy-Littlewood-Pólya Inequalities). Let  $Y \in B^+(\Sigma)$  and let  $A \in \mathcal{G}$  be such that P(A) > 0 and X(A) is a Borel set. Let  $\widetilde{Y}_P$  be the nondecreasing P-rearrangement of Y with respect to X, and let  $\widetilde{Y}_{A,P}$  be the nondecreasing P-rearrangement of Y with respect to X on A. If L is supermodular, then:

- (1)  $\int L(X,Y) dP \leq \int L(X,\widetilde{Y}_P) dP$ , and if L is strictly supermodular then equality holds if and only if  $Y = \widetilde{Y}_P$ , P-a.s., and,
- (2)  $\int_A L(X,Y) dP \leq \int_A L(X,\widetilde{Y}_{A,P}) dP$

provided the integrals exist (i.e. they are not of the form  $\infty - \infty$ ).

**Lemma B.9.** Let  $Y \in B^+(\Sigma)$  and let  $A \in \mathcal{G}$  be such that P(A) > 0 and X(A) is a Borel set. Let  $\widetilde{Y}_P$  be the nondecreasing P-rearrangement of Y with respect to X, and let  $\widetilde{Y}_{A,P}$  be the nondecreasing P-rearrangement of Y with respect to X on A. Then the following hold:

- (1) If  $0 \le Y \le X$ , P-a.s., then  $0 \le \widetilde{Y}_P \le X$ ; and,
- (2) If  $0 \le Y \le X$ , P-a.s. on A, then  $0 \le \widetilde{Y}_{A,P} \le X$ , P-a.s. on A.

#### APPENDIX C. PROOF OF THEOREM 3.1

First note that by the assumption that X is a continuous random variable for the probability measure P, it follows that the Borel probability measure  $P \circ X^{-1}$  is nonatomic. Now, suppose that  $\mathcal{H} = \left\{Y \in B\left(\Sigma\right) \;\middle|\; 0 \leqslant Y \leqslant X \text{ and } \rho\left(Y\right) \leqslant R\right\} \neq \varnothing, \rho \text{ is } (P,X)\text{-vigilant and has the Weak DC-Property, the mapping <math>\mathcal{U}: \mathbb{R}^2 \to \mathbb{R}$  is supermodular, and the mapping  $\mathcal{E}: B^+\left(\Sigma\right) \to B\left(\Sigma\right)$  defined by  $\mathcal{E}\left(Y\right) = \mathcal{U}\left(X,Y\right)$  is uniformly bounded and sequentially continuous in the topology of pointwise convergence.

**Lemma C.1.** For each  $Y \in \mathcal{H}$  there is  $Y^* \in \mathcal{H}$  such that:

- (1)  $Y^*$  is comonotonic with X, i.e.  $Y^*$  is of the form  $I^* \circ X$  where  $I^* : [0, M] \to [0, M]$  is nondecreasing;
- (2)  $\int \mathcal{U}(X, Y^*) dP \geqslant \int \mathcal{U}(X, Y) dP$ ; and,
- (3)  $\rho(Y^*) \leqslant \rho(Y)$ .

Proof. By Assumption,  $\mathcal{H} \neq \emptyset$ . Choose any  $Y = I \circ X \in \mathcal{H}$ , and let  $Y^* := \widetilde{Y}_P$ , where  $\widetilde{Y}_P$  denotes the nondecreasing P-rearrangement of Y with respect to X. Then (i)  $Y^* = \widetilde{I} \circ X$  where  $\widetilde{I} : [0, M] \to [0, M]$  is nondecreasing, bounded, and Borel-measurable; and, (ii)  $0 \leqslant Y^* \leqslant X$ , by Lemma B.9. Furthermore, since  $\rho$  is (P, X)-vigilant, it follows that  $\rho(Y^*) \leqslant \rho(Y)$ , by definition of (P, X)-vigilance. But  $\rho(Y) \leqslant R$  since  $Y \in \mathcal{H}$ . Therefore,  $\rho(Y^*) \leqslant R$ . Thus,  $Y^* \in \mathcal{H}$  is comonotonic with X. Moreover, since the function  $\mathcal{U}$  is supermodular, it follows from Lemma B.8 that  $\int \mathcal{U}(X, Y^*) dP \geqslant \int \mathcal{U}(X, Y) dP$ .

Now, let  $\mathcal{H}^{\uparrow}$  denote the collection of all elements of  $\mathcal{H}$  that are comonotonic with X. Then  $\mathcal{H}^{\uparrow} \neq \emptyset$ , by Lemma C.1. Also, by Lemma C.1, one can choose a maximizing sequence  $\{Y_n\}_n$  in  $\mathcal{H}^{\uparrow}$  for Problem (3.1). That is,  $\lim_{n \to +\infty} \int \mathcal{U}(X, Y_n) \ dP = N := \sup_{Y \in B^+(\Sigma)} \left\{ \int \mathcal{U}(X, Y) \ dP \right\} < +\infty$ . Since  $0 \leqslant Y_n \leqslant X \leqslant M := \|X\|_{\sup}$ , the sequence  $\{Y_n\}_n$  is uniformly bounded. Moreover, for each  $n \geqslant 1$  one has  $Y_n = I_n \circ X$ , with  $I_n : [0, M] \to [0, M]$ . Consequently, the sequence  $\{I_n\}_n$  is a uniformly bounded sequence of nondecreasing Borel-measurable functions. Thus, by Lemma A.1, there is a nondecreasing function  $I^* : [0, M] \to [0, M]$  and a subsequence  $\{I_m\}_m$  of  $\{I_n\}_n$  such that  $\{I_m\}_m$  converges pointwise on [0, M] to  $I^*$ . Hence,  $I^*$  is also Borel-measurable, and so  $Y^* := I^* \circ X \in B^+(\Sigma)$  is such that  $0 \leqslant Y^* \leqslant X$ . Moreover, the sequence  $\{Y_m\}_m$ , defined by  $Y_m = I_m \circ X$ , converges pointwise to  $Y^*$ . Thus, by the assumption that  $\rho$  has the  $Weak\ DC$ -Property,  $Y^* \in \mathcal{H}^{\uparrow}$ . Now, by the assumption of uniform boundedness and sequentially continuity of the map  $\mathcal{U}(X,.)$  in the topology of pointwise convergence, and by Lebesgue's Dominated Convergence Theorem [1, Theorem 11.21] one has

$$\int \mathcal{U}(X, Y^*) \ dP = \lim_{m \to +\infty} \int \mathcal{U}(X, Y_m) \ dP = \lim_{n \to +\infty} \int \mathcal{U}(X, Y_n) \ dP = N$$

Hence  $Y^*$  solves Problem (3.1). The P-a.s. uniqueness of  $Y^*$  is a consequence of Proposition B.3. This concludes the proof of Theorem 3.1.

### APPENDIX D. PROOF OF LEMMA 4.1

Let  $\zeta := P \circ X^{-1}$  be the image measure of P under X. By the assumption that X is a continuous random variable for the probability measure P, the Borel probability measure Q is nonatomic. One can then define the P-a.s. unique nondecreasing P-rearrangement of Q with respect to Q as in Appendix B. Since Q is a nondecreasing function of Q and Q-equimeasurable with Q it follows from the Q-a.s. uniqueness of the equimeasurable nondecreasing Q-rearrangement (see Proposition B.3) that Q is nonatomic, it follows that Q is a uniform distribution over Q is nonatomic, it follows that Q is a uniform distribution over Q in [21, Lemma A.21], that is, Q is Q if Q is Q if Q is a uniform distribution over Q is Q if Q is Q if Q is nonatomic, it follows that Q is Q if Q is Q if Q is Q if Q is nonatomic, it follows that Q is Q if Q is Q if Q is nonatomic, it follows that Q is Q if Q is Q if Q is Q is Q if Q is nonatomic, it follows that Q is Q if Q is Q if Q is Q is Q in Q in Q is Q if Q is Q is Q in Q in Q in Q is Q in Q is Q in Q

Now, let  $Y \in \mathcal{H}$  be given, and let  $\widetilde{Y}$  denote the nondecreasing P-rearrangement of Y with respect to X. Since  $Y \in B^+$  ( $\Sigma$ ), it can be written as  $\phi \circ X$  for some nonnegative Borel-measurable and bounded map  $\phi$  on X (S). Moreover, since  $0 \le Y \le X$ ,  $\phi$  is a mapping of [0, M] into [0, M]. Since,  $\zeta$  is nonatomic (by assumption), one can define the mapping  $\widetilde{\phi} : [0, M] \to [0, M]$  as in Appendix B (see equation (B.2) on p. 18) to be the nondecreasing  $\zeta$ -rearrangement of  $\phi$ , that is,

$$\widetilde{\phi}\left(t\right):=\inf\left\{ z\in\mathbb{R}^{+}\ \middle|\ \zeta\big(\left\{ x\in\left[0,M\right]:\phi\left(x\right)\leqslant z\right\} \right)\geqslant\zeta\big(\left[0,t\right]\big)\right\}$$

Then, as in Appendix B,  $\widetilde{Y} = \widetilde{\phi} \circ X$ . Therefore, for each  $s_0 \in S$ ,

$$\widetilde{Y}\left(s_{0}\right) = \widetilde{\phi}\left(X\left(s_{0}\right)\right) = \inf\left\{z \in \mathbb{R}^{+} \mid \zeta\left(\left\{x \in \left[0, M\right] : \phi\left(x\right) \leqslant z\right\}\right) \geqslant \zeta\left(\left[0, X\left(s_{0}\right)\right]\right)\right\}$$

However, for each  $s_0 \in S$ ,

$$\zeta([0, X(s_0)]) = P \circ X^{-1}([0, X(s_0)]) = F_X(X(s_0)) := F_X(X)(s_0)$$

Moreover,

$$\zeta(\{x \in [0, M] : \phi(x) \le z\}) = P \circ X^{-1}(\{x \in [0, M] : \phi(x) \le z\})$$
$$= P(\{s \in S : \phi(X(s)) \le z\}) = F_Y(z)$$

Consequently, for each  $s_0 \in S$ ,

$$\widetilde{Y}\left(s_{0}\right)=\inf\left\{z\in\mathbb{R}^{+}\mid F_{Y}\left(z\right)\geqslant F_{X}\left(X\right)\left(s_{0}\right)\right\}=F_{Y}^{-1}\left(F_{X}\left(X\left(s_{0}\right)\right)\right):=F_{Y}^{-1}\left(F_{X}\left(X\right)\right)\left(s_{0}\right)$$

That is,

$$\widetilde{Y} = F_V^{-1} \left( F_X \left( X \right) \right)$$

where  $F_Y^{-1}$  is the left-continuous inverse of  $F_Y$ , as defined in equation (4.1). Therefore, the function  $Y^* = F_Y^{-1}(F_X(X))$  coincides with the function  $\widetilde{Y}$  (he nondecreasing *P*-rearrangement of *Y* with respect to *X*).

Consequently, it follows that  $Y^*$  is comonotonic with X. Moreover, by Lemma B.9, it follows that  $0 \le Y^* \le X$ . Also, as in the proof of Lemma C.1, one has  $\int \mathcal{U}(X,Y^*) dP \ge \int \mathcal{U}(X,Y) dP$  (by supermodularity of  $\mathcal{U}$ ),  $\rho(Y^*) \le \rho(Y)$  (by vigilance of  $\rho$ ), and  $Y^* \in \mathcal{H}$ .

#### APPENDIX E. A USEFUL LEMMA

**Lemma E.1.** Suppose that Assumption 4.2, Assumption 5.1, Assumption 5.2, and Assumption 5.6 hold. Let  $\psi$  be the function of the parameter  $\lambda \geq 0$  defined by

$$\psi(\lambda) := \int_{0}^{1} T'(1-t) \max \left[0, \min\left\{F_{X}^{-1}(t), (u')^{-1}(\lambda T'(1-t)) + F_{X}^{-1}(t) - W_{0}\right\}\right] dt$$

Then there exists a  $\lambda^* \ge 0$  such that  $\psi(\lambda^*) = R$ .

*Proof.* Since X is bounded and since  $F_X^{-1}$  is nondecreasing, it follows that for each  $t \in [0,1]$ ,

$$\min \left\{ F_X^{-1}(t), \left( u' \right)^{-1} \left( \lambda T'(1-t) \right) + F_X^{-1}(t) - W_0 \right\} \leqslant F_X^{-1}(t) \leqslant F_X^{-1}(1) \leqslant \|X\|_{sup} < +\infty.$$

Moreover, since T is convex and increasing, T' is nondecreasing and nonnegative, and so for each  $t \in [0,1]$ ,  $0 \le T'(1-t) \le T'(1)$ . But  $T'(1) < +\infty$ , by Assumption 5.6. Hence, for each  $t \in [0,1]$ ,

$$\min \left\{ F_X^{-1}(t), \left(u'\right)^{-1} \left(\lambda T'(1-t)\right) + F_X^{-1}(t) - W_0 \right\} T'(1-t) \leqslant F_X^{-1}(1) T'(1) < +\infty$$

Hence, Lebesgue's Dominated Convergence Theorem [1, Theorem 11.21] implies that  $\psi$  is a continuous function of  $\lambda$ . Moreover,  $\psi$  is a noincreasing function of  $\lambda$  (by concavity of u, convexity of T, and monotonicity of the Lebesgue integral).

Now, Assumption 5.1 implies that

$$\psi(0) = \widehat{\int} X \ dT \circ P = \int_0^1 T'(1-t) F_X^{-1}(t) \ dt$$

and that

$$\lim_{\lambda \to +\infty} \psi(\lambda) = \int_0^1 T'(1-t) \max \left[ 0, \min \left\{ F_X^{-1}(t), F_X^{-1}(t) - W_0 \right\} \right] dt$$
$$= \int_0^1 T'(1-t) \max \left[ 0, F_X^{-1}(t) - W_0 \right] dt$$

Furthermore, by Assumption 5.2,  $F_X(W_0) = 1$ . Therefore, for all  $t \in (0,1)$ ,  $F_X^{-1}(t) \leq W_0$ , and

$$\lim_{\lambda \to +\infty} \psi\left(\lambda\right) = 0$$

Consequently (recall Remark 5.3),

$$0 = \lim_{\lambda \to +\infty} \psi(\lambda) \leqslant R \leqslant \widehat{\int} X \ dT \circ P = \int_{0}^{1} T'(1-t) F_{X}^{-1}(t) \ dt = \psi(0)$$

Hence, by the Intermediate Value Theorem [50, Theorem 4.23], there exists a  $\lambda^* \ge 0$  such that  $\psi(\lambda^*) = R$ .

#### References

- [1] C.D. ALIPRANTIS and K.C. BORDER. Infinite Dimensional Analysis 3<sup>rd</sup> edition. Springer-Verlag, 2006.
- [2] M. AMARANTE, M. GHOSSOUB, and E.S. PHELPS. The Entrepreneurial Economy I: Contracting under Knightian Uncertainty. Columbia University, Center on Capitalism and Society, Working Paper No. 68 (April 2011).
- [3] K.J. Arrow. Essays in the Theory of Risk-Bearing. Chicago: Markham Publishing Company, 1971.
- [4] P. Artzner, F. Delbaen, J.M. Eber, and D. Heath. Coherent Measures of Risk. *Mathematical Financ*, 9(3):203–228, 1999.
- [5] S. Athey. Monotone Comparative Statics under Uncertainty. Quarterly Journal of Economics, 117(1):187–223, 2002.
- [6] A. Balbás. Mathematical Methods in Modern Risk Measurement: A Survey. Revista de la Real Academia de Ciencias Exactas, Físicas y Naturales. Serie A: Matemáticas (RACSAM), 101(2):205–219, 2007.
- [7] C. Bennett and R. Sharpley. Interpolation of Operators. Academic Press, 1988.

- [8] K.H. BORCH. The Mathematical Theory of Insurance: An Annotated Selection of Papers on Insurance Published 1960-1972. Lexington Books, 1974.
- [9] G. CARLIER and R.A. DANA. Law Invariant Concave Utility Functions and Optimization Problems with Monotonicity and Comonotonicity Constraints. Statistics & Decisions, 24(1):127–152, 2006.
- [10] G. CARLIER and R.A. DANA. Optimal Demand for Contingent Claims when Agents Have Law Invariant Utilities. Mathematical Finance, 21(2):169–201, 2011.
- [11] N.L. CAROTHERS. Real Analysis. Cambridge University Press, 2000.
- [12] K.M. CHONG and N.M. RICE. Equimeasurable Rearrangements of Functions. Queens Papers in Pure and Applied Mathematics, 28, 1971.
- [13] D.L. Cohn. Measure Theory. Birkhauser, 1980.
- [14] J. CVITANIĆ, W. SCHACHERMAYER, and H. WANG. Utility Maximization in Incomplete Markets with Random Endowment. Finance and Stochastics, 5(2):259–272, 2001.
- [15] R.A. DANA. A Representation Result for Concave Schur Concave Functions. Mathematical Finance, 15(4):613–634, 2005.
- [16] F. Delbaen. Coherent Risk Measures on General Probability Spaces. Advances in finance and stochastics. Essays in Honour of Dieter Sondermann, pages 1–37, 2002.
- [17] J. Dieudonné. Foundations of Modern Analysis. Academic press, 1969.
- [18] P.H. Dybvig. Distributional Analysis of Portfolio Choice. Journal of Business, 61(3):369–393, 1988.
- [19] P.H. DYBVIG. Inefficient Dynamic Portfolio Strategies or How to Throw Away a Million Dollars in the Stock Market. Review of Financial Studies, 1(1):67–88, 1988.
- [20] J.B. EPPERSON. A Class of Monotone Decreasing Rearrangements. Journal of Mathematical Analysis and Applications, 150(1):224–236, 1990.
- [21] H. FÖLLMER and A. SCHIED. Stochastic Finance: An Introduction in Discrete Time 3<sup>rd</sup> ed. Walter de Gruyter, 2011.
- [22] D. GALE and M. HELLWIG. Incentive-Compatible Debt Contracts: The One-Period Problem. The Review of Economic Studies, 52(4):647–663, 1985.
- [23] M. GHOSSOUB. Belief Heterogeneity in the Arrow-Borch-Raviv Insurance Model. mimeo (2011).
- [24] M. Ghossoub. Monotone Equimeasurable Rarrangements with Non-Additive Probabilities. mimeo (2011).
- [25] M. Ghossoub. Supplement to 'Belief Heterogeneity in the Arrow-Borch-Raviv Insurance Model'. mimeo (2011).
- [26] M. GHOSSOUB. Contracting under Heterogeneous Beliefs. PhD thesis, University of Waterloo Dept. of Statistics & Actuarial Science, 2011.
- [27] S.J. GROSSMAN and O.D. HART. An Analysis of the Principal-Agent Problem. Econometrica, 51(1):7–45, 1983.
- [28] G.H. HARDY, J.E. LITTLEWOOD, and G. PÓLYA. *Inequalities*. Cambridge University Press, Cambridge. Reprint of the 1952 edition, 1988.
- [29] X.D. HE and X.Y. ZHOU. Portfolio Choice via Quantiles. Mathematical Finance, 21(2):203–231, 2011.
- [30] B. HOLMSTROM. Moral Hazard and Observability. The Bell Journal of Economics, 10(1):74–91, 1979.
- [31] J. Hugonnier and D. Kramkov. Optimal Investment with Random Endowments in Incomplete Markets. The Annals of Applied Probability, 14(2):845–864, 2004.
- [32] K. INADA. On a Two-Sector Model of Economic Growth: Comments and a Generalization. The Review of Economic Studies, 30(2):119–127, 1963.
- [33] H. JIN and X.Y. ZHOU. Greed, Leverage, and Potential Losses: A Prospect Theory Perspective. Mathematical Finance, Forthcoming.
- [34] H. JIN and X.Y. ZHOU. Behavioral Portfolio Selection in Continous Time. Mathematical Finance, 18(3):385–426, 2008.
- [35] H. Jin and X.Y. Zhou. Erratum to "Behavioral Portfolio Selection in Continuous Time". *Mathematical Finance*, 20(3):521–525, 2010.
- [36] D. KRAMKOV and W. SCHACHERMAYER. The Asymptotic Elasticity of Utility Functions and Optimal Investment in Incomplete Markets. The Annals of Applied Probability, pages 904–950, 1999.
- [37] D. Kramkov and W. Schachermayer. Necessary and Sufficient Conditions in the Problem of Optimal Investment in Incomplete Markets. *The Annals of Applied Probability*, 13(4):1504–1516, 2003.
- [38] W.A.J. Luxemburg. Rearrangement Invariant Banach Function Spaces. In *Queens Papers in Pure and Applied Mathematics, vol. 10: Proceeding of the Symposium in Analysis*, pages 83–144. Queen's University, Kingston, Ontario, 1967.

- [39] M. MARINACCI and L. MONTRUCCHIO. Introduction to the Mathematics of Ambiguity. In I. Gilboa (ed.), Uncertainty in Economic Theory: Essays in Honor of David Schmeidlers 65th Birthday. Routledge, London, 2004.
- [40] R. MERTON. Lifetime Portfolio Selection under Uncertainty: The Continuous-Time Case. The Review of Economics and Statistics, 51(3):247–257, 1969.
- [41] R. MERTON. Optimum Consumption and Portfolio Rules in a Continuous-Time Model. Journal of Economic Theory, 3:373–413, 1971.
- [42] P. MILGROM and J. ROBERTS. The Economics of Modern Manufacturing: Technology, Strategy, and Organization. The American Economic Review, 80(3):511–528, 1990.
- [43] P. MILGROM and C. SHANNON. Monotone Comparative Statics. Econometrica, 62(1):157–180, 1994.
- [44] J.A. MIRRLEES. The Theory of Moral Hazard and Unobservable Behaviour: Part I. Review of Economic Studies, 66(1):3–21, 1999.
- [45] M.P. OWEN and G. ŽITKOVIĆ. Optimal Investment with an Unbounded Random Endowment and Utility-Based Pricing. Mathematical Finance, 19(1):129–159, 2009.
- [46] F.H. Page. The Existence of Optimal Contracts in the Principal-Agent model. *Journal of Mathematical Economics*, 16(2):157–167, 1987.
- [47] E. Pap. Null-Additive Set Functions. Kluwer Academic Publishers, 1995.
- [48] A. RAVIV. The Design of an Optimal Insurance Policy. The American Economic Review, 69(1):84–96, 1979.
- [49] W.P. ROGERSON. The First-Order Approach to Principal-Agent Problems. Econometrica, 53(6):1357–1367, 1985.
- [50] W. Rudin. Principles of Mathematical Analysis 3<sup>rd</sup> ed. New York: McGraw-Hill Book Company, 1976.
- [51] W. SCHACHERMAYER. Optimal Investment in Incomplete Markets When Wealth May Become Negative. The Annals of Applied Probability, 11(3):694–734, 2001.
- [52] A. Schied. On the Neyman-Pearson Problem for Law-Invariant Risk Measures and Robust Utility Functionals. Annals of Applied Probability, 14(3):1398–1423, 2004.
- [53] A. Schied. Optimal Investments for Robust Utility Functionals in Complete Market Models. Mathematics of Operations Research, 30(3):750-764, 2005.
- [54] A. Schied. Risk Measures and Robust Optimization Problems. Stochastic Models, 22(4):753–831, 2006.
- [55] A. Schied. Optimal Investments for Risk- and Ambiguity-Averse Preferences: A Duality Approach. Finance and Stochastics, 11(1):107–129, 2007.
- [56] A. Schied and C.T. Wu. Duality Theory for Optimal Investments under Model Uncertainty. Statistics & Decisions, 23(3):199–217, 2005.
- [57] D.M. TOPKIS. Minimizing a Submodular Function on a Lattice. Operations Research, 26(2):305–321, 1978.
- [58] D.M. TOPKIS. Supermodularity and Complementarity. Princeton University Press, 1998.
- [59] R. TOWNSEND. Optimal Contracts and Competitive Markets with Costly State Verification. Journal of Economic Theory, 21(2):265–293, 1979.
- [60] X. VIVES. Oligopoly Pricing: Old Ideas and New Tools. MIT Press, 2001.
- [61] S.D. WILLIAMSON. Costly Monitoring, Financial Intermediation, and Equilibrium Credit Rationing. *Journal of Monetary Economics*, 18(2):159–179, 1986.
- [62] S. Zhang, H. Jin, and X.Y. Zhou. Behavioral Portfolio Selection with Loss Control. Acta Mathematica Sinica, English Series, 27(2):255–274, 2011.

Université de Montréal – Département de Sciences Économiques – C.P. 6128, succursale Centreville – Montréal, QC, H3C 3J7 – Canada

 $E ext{-}mail\ address: mario.ghossoub@umontreal.ca}$